\begin{table}%t5
\caption{Properties of known low-mass TTS in the EPIC field of view. Column~[2] is the source name (with  prefix~"SZ'' from Schwartz \cite{Schw77}, "TTS'' from Krautter et~al. \cite{Krau97}, and "Par-Lup'' from Comeron et~al. \cite{Come03}). Column~[3] gives the spectral type. Columns~[10] and~[11] provide the the H$_{\rm \alpha}$ equivalent width and the $\Delta$$(H-K)$~color excess from Hughes et~al. (\cite{Hugh94}), Wichmann et~al. (\cite{Wich97b}), and Comeron et~al. (\cite{Come03}).}\label{tab5}
\small%\centerline
{
\begin{tabular}{c c c c c c c c c c c}
\hline
\hline
S & Id & Sp. & $L_{\rm X}$                &  $L_{\rm bol}$  &  $L_{\rm
X}$/$L_{\rm bol}$ & $T_{\rm eff}$ & $M$ & Age & $W_{\rm H_{\alpha}}$ & $\Delta$$(H-K)$ \\
  &    &     & (10$^{28}$ erg s$^{-1}$)   & ($L_\odot$) & (10$^{-3}$) & (K) &
($M_\odot$) & (10$^{6}$ yr) & (\AA) & \\
$[$1$]$ & [2] & [3] & [4] & [5] & [6] & [7] & [8] & [9] & [10] & [11] \\
\hline
8 &      SZ 94 & M4    &   8.3 $\pm$ 0.7  & 0.07 & 0.31 & 3150 & 0.18 & 2.6 & 7.3 & --0.01 \\
10 &     SZ 95 & M1.5  &  13.0 $\pm$ 1.6 & 0.18 & 0.19 & 3570 & 0.53 & 5.7 & 10.2 & 0.03 \\
13 &    TTS 106 & M0   & 128.5 $\pm$ 3.2 & 0.38 & 0.88 & 3800 & 0.86 & 6.4 & & 0.02 \\
22 &    TTS 107 & K6   &  26.6 $\pm$ 1.3 & 0.28 & 0.25 & 4260 & 0.88 & 28 & & 0.03 \\
24 &     SZ 96 & M1.5  & 278.0 $\pm$ 4.0 & 0.29 & 2.51 & 3570 & 0.55 & 3.6 & 1.0 & 0.20 \\          
28 & Par-Lup3-1 & M7.5 &   7.9 $\pm$ 0.6 & 0.05 & 0.41 & 2760 & 0.04$^{*}$ & $<$1$^{**}$ & 12 & \\
   & Par-Lup3-1/cc1 &  &                 &      &      &      & & &  & \\
36 &     SZ 98 & K8    &  43.5 $\pm$ 1.8 & 0.62 & 0.18 & 3800 & 0.90 & 2.9 & 29.1 & 0.55 \\
37 &    TTS 111 & M3.5 &   6.6 $\pm$ 1.1 & 0.06 & 0.29 & 3950 & 0.60 & 147 & & 0.03 \\
41 &     SZ 100 & M5   &  18.5 $\pm$ 1.2 & 0.12 & 0.40 & 3000 & 0.10$^{*}$ & $<$1$^{**}$ & 21.4 & 0.14 \\
44 &    TTS 110 & M1.5 &  51.9 $\pm$ 1.7 & 0.14 & 0.97 & 3570 & 0.52 & 8.5 & & 0.03 \\
45 &     SZ 101 & M4   &  20.9 $\pm$ 1.3 & 0.19 & 0.29 & 3150 & 0.17 & 1.0 & 26.0 & --0.08 \\
50 &     SZ 102 & K0   &   3.8 $\pm$ 0.7 & 0.01 & 0.86 & & & & 377.4 & 1.01 \\
62 & Par-Lup3-2 & M6   &   4.3 $\pm$ 0.5 & 0.20 & 0.06 & 2970 & 0.09$^{*}$ & $<$1$^{**}$ & 6.5 & \\
73 &     SZ 107 & M5.5 &   4.0 $\pm$ 0.5 & 0.07 & 0.15 & 2900 & 0.07$^{*}$ & $<$1$^{**}$ & 16.1 & --0.02 \\
74 &     SZ 108 & M0.5 &  17.8 $\pm$ 0.9 & 0.44 & 0.11 & 3650 & 0.69 & 3.1 & 0.5 & 0.11 \\
78 &     SZ 109 & M5.5 &   7.0 $\pm$ 0.7 & 0.07 & 0.26 & 2900 & 0.06$^{*}$ & $<$1$^{**}$ & 21.7 & 0.06 \\
80 & Par-Lup3-3 & M4.5 &  27.7 $\pm$ 2.2 & 0.63 & 0.11 & 3420 & 0.53 & 1.0 & 19 & \\
83 &     SZ 110 & M2   &  14.6 $\pm$ 1.4 & 0.17 & 0.22 & 3500 & 0.45 & 5.1 & 60.1 & 0.13 \\
87 &    TTS 116 & K2   & 318.2 $\pm$ 5.7 & 0.76 & 1.09 & 4900 & 1.04 & 26 & & 0.06 \\
88 &     SZ 112 & M3   &  18.3 $\pm$ 1.4 & 0.14 & 0.34 & 3150 & 0.19 & 1.4 & 46.7 & 0.16 \\
92 &     SZ 114 & M2   &  90.4 $\pm$ 2.9 & 0.22 & 1.07 & 3150 & 0.17 & 1.0 & 33.0 & 0.13 \\
94 &     SZ 115 & M3   &   3.1 $\pm$ 0.5 & 0.08 & 0.10 & 3150 & 0.19 & 2.9 & 7.1 & 0.05 \\
   &     SZ  97 & M3   &    $<$3.8 & 0.13 & $<$0.08 & 3350 & 0.31 & 3.2 & 58.2 & 0.00 \\
   &     SZ  99 & M3.5 &    $<$4.7 & 0.08 & $<$0.15 & 3250 & 0.23 & 4.5 & 49.8 & 0.02 \\
   &     SZ 103 & M4   &    $<$2.7 & 0.06 & $<$0.12 & 3150 & 0.18 & 3.4 & 33.1 & 0.21 \\
   &     SZ 104 & M5.5 &    $<$2.7 & 0.05 & $<$0.14 & 3000 & 0.10 & 2.1 & 13.3 & 0.12 \\
   &     SZ 106 & K8   &    $<$3.5 & 0.03 & $<$0.30 & 3800 & 0.50 & $>$250 & 81.7 & 0.72 \\ 
   &     SZ 113 & M1.5 &    $<$4.7 & 0.03 & $<$0.41 & 3150 & 0.14 & 5.7 & 160.1 & 0.12 \\
   & Par-Lup3-4 & M5e  &    $<$13.8 & 0.006 & $<$6.0 & 3125 & 0.11 & 35 & 410 & \\
\hline
\end{tabular}}
\medskip
$^{*}$ Uncertain mass estimate; $^{**}$ uncertain age estimate.
\end{table}